Author's response to reviews

Title: A population-based study of human immunodeficiency virus in south India reveals major differences from sentinel surveillance-based estimates

Authors:

Lalit Dandona (lalit.dandona@gmail.com)
Vemu Lakshmi (lgorthi@hotmail.com)
Talasila Sudha (s_talasila@hotmail.com)
G Anil Kumar (akgkerala@yahoo.co.in)
Rakhi Dandona (rakhi.dandona@gmail.com)

Version: 3 Date: 6 December 2006

Author's response to reviews:

6 December 2006

Dear Editors,

The checked copy-edited version of the manuscript is uploaded.

Kind regards,
Lalit Dandona